Author's response to reviews

Title: The correlation of WWOX, RUNX2, and VEGFA protein expression in human osteosarcoma

Authors:

Jilong Yang (yangjilong@timuch.com)
Linru Zhao (zhao_linru@126.com)
Wei Tian (tianweiyh@yahoo.com.cn)
Hong Zheng (zhengh64@yahoo.com.cn)
Guowen Wang (wgwhrb@yahoo.com.cn)
Kexin Chen (chenkexin1963@yahoo.com)
Zhichao Liao (zhichaoli@hotmail.com)

Version: 4 Date: 8 December 2013

Author's response to reviews: see over
Dear Prof. Tim Sands,

I would like to thank you to accept, in principle, our revised manuscript entitled “The correlation of WWOX, RUNX2, and VEGFA protein expression in human osteosarcoma” for publication in BMC Medical Genomics. The new information is marked as blue text in the revised manuscript. Our detailed response is listed below.

Tim Sands
Executive Editor
Peer review of your manuscript (above) is now complete, and we are delighted, in principle, to accept the manuscript for publication in BMC Medical Genomics. However, before acceptance, we will require you to include the name of the institutional review board that approved the study.

Response:
We have added the name of Institutional Review Board (IRB).

After completion of the modification, I hope you will find the revised version suitable for publication in BMC Medical Genomics.

Please do not hesitate to contact me if you need additional information.

Sincerely yours,

Jilong Yang, M.D., Ph.D.